Author’s response to reviews

Title: Salivary extracellular vesicle-associated miRNAs as potential biomarkers in oral squamous cell carcinoma.

Authors:

Chiara Gai (chiara.gai@unito.it)
Francesco Camussi (francescocamussi@yahoo.it)
Roberto Broccoletti (roberto.broccoletti@unito.it)
Alessio Gambino (alessio.gambino@unito.it)
Marco Cabras (cabrasmarco300@gmail.com)
Luca Molinaro (luca.molinaro@unito.it)
Stefano Carossa (stefano.carossa@unito.it)
Giovanni Camussi (giovanni.camussi@unito.it)
Paolo Arduino (paologiacomo.arduino@unito.it)

Version: 4 Date: 06 Apr 2018

Author’s response to reviews:

To the editor of BMC Cancer,

Enclosed please find our manuscript BCAN-D-17-00903R3 revised according to Editor suggestions and a point by point response to Editor comments.

1) As you suggested, we revised the paragraph Authors' contributions in Declarations (page 24-25) section completing the contribution of the authors AG, MC, SC.

2) The manuscript has been uploaded as a single, final, clean version.